Author's response to reviews

Title: Suicidal intention, psychosocial factors and referral to further treatment - a one-year cross-sectional study of self-poisonings

Authors:

Mari A Bjornaas (mabjornaas@gmail.com)
Knut E Hovda (kehovda@yahoo.no)
Fridtjof Heyerdahl (fridtjof.heyerdahl@medisin.uio.no)
Karina Skog (karina.skog@akersykehus.no)
Per Drottning (per.drottning@lds.no)
Anders Opdahl (anders.opdahl@medisin.uio.no)
Dag Jacobsen (daja@uus.no)
Oivind Ekeberg (oivind.ekeberg@uus.no)

Version: 4 Date: 15 July 2010

Author's response to reviews: see over
Oslo, July 15, 2010

Dear Editor,

Thank you for your acceptance of our manuscript. We have now done the formatting changes as requested, see below.

Please contact me if further changes are needed.

Yours sincerely,

Mari Asphjell Bjørnaas
MD, PhD
Department of Acute Medicine
Oslo University Hospital, Ulleval
N-0407 Oslo

E-mail: mabjornaas@gmail.com

Formatting changes to be made:
-----------------------------------
**Major revisions** (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name, Initial, Last name' (e.g. Joe F Bloggs)

*MB: Done.*

**Minor revisions** (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

*MB: Done.*

Tables- please remove the visible vertical lines from your tables

*MB: Done*

**Typography:** Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

*MB: Done.*